Author's response to reviews

Title: Rationale, design, and results of the first screening round of a comprehensive, register-based, Chlamydia screening implementation programme in the Netherlands

Authors:

Jan EAM van Bergen (jvanbergen@soaaids.nl)
Johannes SA Fennema (hfennema@ggd.amsterdam.nl)
Ingrid VF van den Broek (ingrid.van.den.broek@rivm.nl)
Elfi EHG Brouwers (elfi.brouwerd@ggdzl.nl)
Eva M de Feijter (evamdef@hotmail.com)
Christian JPA Hoebe (christia.hoebe@ggdzl.nl)
Rik H Koekenbier (rkoekenbier@ggd.amsterdam.nl)
Eline LM Op de Coul (eline.op.de.coul@rivm.nl)
Sander M van Ravesteijn (vanravesteijns@ggd.rotterdam.nl)
Hannelore M Götz (GotzH@ggd.rotterdam.nl)

Version: 5  Date: 4 October 2010

Author's response to reviews:

Dear Editor and production team,

thank you for your carefull revision of the document. Indeed I encountered 2 small differences in authorlist, which are revised in the submission system. They now match with the manuscript details.

The references are in numerical order and now in consecutive numbers and between brackes. The reference list has been adapted accordingly to some changes in the numerical order.

kind regards.

Jan van Bergen